Author's response to reviews

Title: Difference of clinical features in childhood Mycoplasma pneumoniae pneumonia

Authors:

You-Sook Youn (vanco77@hanmail.net)
Kyung-Yil Lee (leekyungyil@catholic.ac.kr)
Ja-Young Hwang (HJY@catholic.ac.kr)
Jung-Woo Rhim (benign7@hanmail.net)
Jin-Han Kang (kjhan@catholic.ac.kr)
Joon-Sung Lee (leejs@catholic.ac.kr)
Ji-Chang Kim (jichang71@hotmail.net)

Version: 5 Date: 30 June 2010

Author's response to reviews: see over
Melissa Norton, MD  
Editor-in-chief of *BMC Pediatrics*  
June 30, 2010

Dear Dr Norton

We appreciate your accept of our manuscript. We corrected the manuscript according to the directions in e-mail of the journal; Affiliations, Authors’ contributions, removing of all highlighting in manuscript, rename Methods, and use figure in both the main text and figure legends section.

Thank you in advance for considering our revised manuscript.

Kind regards,  
Professor Kyung-Yil Lee, MD, on behalf of the authors  
Department of Pediatrics, The Catholic University of Korea,  
Daejeon St. Mary's Hospital, 520-2 Daeheung-dong, Jung-gu,  
Daejeon 301-723, Republic of Korea  
Tel: +82-42-220-9541, Fax: +82-42-221-2925  
E-mail: leekyungyil@catholic.ac.kr